Author’s response to reviews

Title: Global prevalence of cardiometabolic risk factors in the military population: a systematic review and meta-analysis

Authors:

Fereshteh Baygi (fb.baygi@gmail.com)
Kimmo Herttua (kherttua@hea.sdu.dk)
Olaf Jensen (ocj@hea.sdu.dk)
Shirin Djalalinia (shdjalalinia@gmail.com)
Armita Ghorabi (armitamahdavi61@gmail.com)
Hamid Asayesh (hasayesh@gmail.com)
mostafa qorbani (mqorbani1379@yahoo.com)

Version: 2 Date: 28 Dec 2019

Author’s response to reviews:

Editor Comments:

1. Please relocate the PRISMA flowchart to be after the References along with the other figures and tables.
   -Answer: Thank you. It was done.

2. Please include an Acknowledgments subsection in your Declarations. The Acknowledgments subsection is your opportunity to acknowledge anyone who contributed towards the article who does not meet the criteria for authorship including anyone who provided professional writing services or materials. If you have no individuals to acknowledge you may write “Not applicable” in this subsection.
   -Answer: Thank you. It was done.

3. Please provide a “List of abbreviations” subsection after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.
   -Answer: Thank you. It was done.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
   -Answer: Thank you. It was done.